Author's response to reviews

Title: Use of RE-AIM to Develop a Multi-Media Facilitation Tool for the Patient-Centered Medical Home

Authors:

Russell E Glasgow (glasgowre@mail.nih.gov)
Perry Dickinson (perry.dickinson@ucdenver.edu)
Lawrence Fisher (fisherl@fcm.ucsf.edu)
Steve Christiansen (steve@intervisionmedia.com)
Deborah J Toobert (deborah@ori.org)
Bruce G Bender (benderb@njc.org)
L Miriam Dickinson (miriam.dickinson@uchsc.edu)
Bonnie Jortberg (bonnie.jortberg@ucdenver.edu)
Paul A Estabrooks (estabrkp@vt.edu)

Version: 4 Date: 6 October 2011

Author's response to reviews: see over
September 26, 2011

Implementation Science Editorial Team

RE: MS 468418525268478 – Glasgow

Dear Editorial Team:

Attached please find the “clean” copy of our manuscript as mentioned above by manuscript number for Implementation Science.

We confirm that we have accepted all the proposed changes by the copy editor.

Sincerely,

Russell E. Glasgow, Ph.D.
Deputy Director for Implementation Science
Division of Cancer Control and Population Sciences
NCI
6130 Executive Blvd., Room 6144
Rockville, MD 20852